# **Coevolution and the Effects of Climate Change on Interacting Species**

# Tobin D. Northfield<sup>\*\*</sup>, Anthony R. Ives

Department of Zoology, University of Wisconsin, Madison, Wisconsin, United States of America

# Abstract

**Background:** Recent studies suggest that environmental changes may tip the balance between interacting species, leading to the extinction of one or more species. While it is recognized that evolution will play a role in determining how environmental changes directly affect species, the interactions among species force us to consider the coevolutionary responses of species to environmental changes.

*Methodology/Principle Findings:* We use simple models of competition, predation, and mutualism to organize and synthesize the ways coevolution modifies species interactions when climatic changes favor one species over another. In cases where species have conflicting interests (i.e., selection for increased interspecific interaction strength on one species is detrimental to the other), we show that coevolution reduces the effects of climate change, leading to smaller changes in abundances and reduced chances of extinction. Conversely, when species have nonconflicting interests (i.e., selection for increased interspecific interaction strength on one species conversely, when species have nonconflicting interests (i.e., selection for increased interspecific interaction strength on one species benefits the other), coevolution increases the effects of climate change.

*Conclusions/Significance:* Coevolution sets up feedback loops that either dampen or amplify the effect of environmental change on species abundances depending on whether coevolution has conflicting or nonconflicting effects on species interactions. Thus, gaining a better understanding of the coevolutionary processes between interacting species is critical for understanding how communities respond to a changing climate. We suggest experimental methods to determine which types of coevolution (conflicting or nonconflicting) drive species interactions, which should lead to better understanding of the effects of coevolution on species adaptation. Conducting these experiments across environmental gradients will test our predictions of the effects of environmental change and coevolution on ecological communities.

Citation: Northfield TD, Ives AR (2013) Coevolution and the Effects of Climate Change on Interacting Species. PLoS Biol 11(10): e1001685. doi:10.1371/journal.pbio.1001685

Academic Editor: Nick H. Barton, Institute of Science and Technology Austria (IST Austria), Austria

Received June 18, 2013; Accepted September 12, 2013; Published October 22, 2013

**Copyright:** © 2013 Northfield, Ives. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: TDN was supported by a USDA Postdoctoral fellowship (#WISW-2010-05109). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: tobin.northfield@jcu.edu.au

¤ Current address: Centre for Tropical Biodiversity and Climate Change, School of Marine and Tropical Biology, James Cook University, Cairns, Queensland, Australia

# Introduction

Climatic changes, or indeed any change in the environment, have the potential to cause the local extinction of species, and to alter community composition and ecosystem functioning [1]. Numerous models have been used to predict how the density and geographical range of species will be affected by climate change, with mixed success [2]. In part, success is limited by the need to understand how changes in the density of one species affect densities of other species through their interactions. For example, reduced pollinator densities resulting from global climate change have led to local extinction of several plant species [3]. Climate change may also have direct effects on the strength of species interactions, and these are sometimes difficult to predict [4]. For example, climatic change has been implicated in mediating extinctions of amphibian species by altering the epidemiology of their pathogens [5]. Thus, the effects of climate change on species densities and extinction risks depend both on the direct effects of climate change on focal species and on the indirect effects acting through interactions among species, making predictions of the net effect of climate change on extinction risk challenging [6–9].

Rapid evolution is also expected to influence species responses to climatic changes [10,11]. In addition to evolution directly driven by the changing climate, coevolution between species may modify species interactions [12,13]. For example, Zhang and Buckling [14] factorially manipulated the environment and a bacterium's ability to coevolve in a bacterium-phage virus system where the environmental changes reduced phage infection. The phage could persist in the presence of a changing environment or when the host was allowed to evolve, but not when both environmental change and host evolution occurred simultaneously. This phage extinction was likely caused by a combination of increased costs associated with the coevolutionary arms race and a reduced effective population size resulting from a deteriorating environment [14]. Thus, coevolution has the potential to alter species interactions to the point of reversing the fate of the

# **Author Summary**

Recent studies suggest that environmental changes may tip the balance between species that interact with each other, leading to the extinction of one or more species. While it is recognized that evolution will alter the way environmental changes directly affect individual species, the interactions between species force us to also consider the evolution of species interactions themselves. We use simple models of competition, predation, and mutualism to evaluate the effect of coevolution on the abundance of interacting species when climatic changes favor one species over another. In cases where the species have conflicting interests (i.e., where selection on one species for increased strength of the interaction is detrimental to the other, such as an organism becoming more aggressive towards competitors), we show that coevolution reduces the effects of climate change, leading to smaller changes in abundances and reduced chances of extinction. Conversely, when the species have nonconflicting interests (i.e., where selection for increased interaction strength on one species benefits the other, such as an organism avoiding competition with other species), coevolution increases the effects of climate change. Thus, gaining a better understanding of the nature of the coevolution between interacting species is critical for understanding how communities respond to a changing climate.

interacting species and is therefore likely to be an important determinant of extinction risk and community composition [15–17].

The theory addressing the evolution of species in direct response to a changing climate is well known in the context of climate change (reviewed by [18]), and there is also a relevant theoretical literature addressing coevolution [19,20]. Coevolutionary analyses of the effects of productivity on coevolving ecological communities give insights into expected community responses to climate-driven changes in densities of particular species. For example, Hochberg and van Baalen [21] used predator-prey coevolution models to show that increased prey productivity can lead to increased defense against predators and a stronger arms race. Similarly, Abrams and Vos [22] demonstrated that in some scenarios increased prey mortality can lead to increased predator density, as prey invest less in predator defense. Indeed, microcosm experiments have demonstrated that increased resource abundance for a prey species can lead to increased prey defense, resulting in lower predator-prey ratios [23,24]. If the cost of predator defense is associated with reduced intraspecific competitive ability, selection against well-defended phenotypes is expected to be strongest when competition is strong and predation weak [25], and several empirical studies have demonstrated that this type of coevolution can drive population dynamics (e.g., [26-28]). Theoretical studies focused on nutrient availability and range expansion have suggested that coevolution of competitors may also alter the effects of climate change on communities. As resources decline, divergent coevolution has the potential to reduce the ratio of interspecific to intraspecific competition, leading to increased coexistence in the presence of low resource availability [29]. In cases where climate change leads to range expansion and sympatric competitor distributions, divergent coevolution can lead to increased coexistence [30].

A frequent conclusion in these and other studies is that coevolution should be stabilizing, reducing changes in population densities of interacting species (e.g., [31,32]). Here, we examine this hypothesis in detail. Our goal is to develop a simple, general theoretical framework to organize and synthesize the ways coevolution could modify the outcome of changing environmental conditions that will likely be pervasive with climate change.

# **Modeling Coevolution**

To evaluate the effects of climate change, we present three coevolutionary models describing competitive, mutualistic, and predator-prey relationships between two species. Spatial structure may influence the effects of climate change on coevolving species [19,20]; intermediate dispersal levels may slow local adaptation by diluting locally adapted genotypes, while low dispersal levels may speed local adaptation by providing advantageous genotypes [33]. In addition, when the climate itself varies across space, intermediate dispersal levels could lead to a geographic mosaic of coevolution where selection pressures and species traits vary across space [34]. Nonetheless, to focus on local adaption, we assumed that each species is represented by a single, panmictic population.

We modeled species interactions in terms of population dynamics: how the density of one species affects the population growth rate of the other. For example, for predation a high density of the predator will lead to a decrease in the population growth rate of the prey, and a high density of the prey will lead to an increase in the population growth rate of the predator; note that this general definition of predator-prey interactions encompasses host-pathogen and plant-herbivore interactions. While there is only a single interaction between species in the model, it is modeled as two parameters, one for the effect of the interaction on each species. Thus, for competitive interactions, one interaction parameter measures the negative effect of the density of the first species on the population growth rate of the second, and another parameter measures the effect of the second species on the population growth rate of the first.

We further assumed that each species has a trait that affects the strength of these interaction parameters. For example, a prev has a defensive trait that simultaneously decreases the negative effect of predation it experiences and decreases the positive effect accrued by the predator; similarly, a predator has an offensive trait that increases the predation rate on prey and increases the benefits obtained by the predator. Note that, in contrast to many models of species coevolution [35-37], we did not assume that there is trait matching in which the strength of interaction depends on a match between the traits values of each species; in our model both species have traits that cause monotonic benefits to the species. These benefits, however, have a cost that is exacted by decreases in the intrinsic rate of increase of the species. For example, a prey might increase its defensive trait and as a consequence suffer a reduced reproduction rate. Finally, we modeled trait evolution using a quantitative genetics approach, so the rate of evolution depends on the strength of selection and the additive genetic variance of the trait, where the additive genetic variance is constant. While this assumption about evolution is unlikely to hold in the long term (when mutations will be needed to maintain genetic variation), under very strong selection (which will cause loss of genetic variation), and for small populations (that lack large initial genetic variation and experience genetic drift), it is a reasonable starting point to investigate the short-term (hundreds of generations) response of species to climate change [38].

We used the models to pose the question: If the environment changes in such a way that the intrinsic rate of increase of one species rises, how will coevolution affect the equilibrium densities of both species? By "equilibrium density" we mean the density that would be obtained if changes in population density occurred on a more rapid time scale than evolution, although as we describe, this assumption gives insight into the case of rapid evolution on the same time scale as changes in density. We made the simplifying assumption that only one of the species experiences a direct change in its intrinsic rate of increase caused by the environmental change; this just makes it easier to separate the evolutionary changes in one species that directly experiences environmentally driven demographic changes from the other species that only responds indirectly through its interactions with the first. There is no loss of generality with this assumption, however, since the net effect of environmental changes to both species would be, to a first approximation, the simple combination of environmental changes to each species separately (Box 1).

There is a rich history of studies that show the effects of environmental change on demographic factors that affect the intrinsic rates of increase of species. For example, higher temperatures often lead to increased development rates in ectotherms, a relationship that is easily quantified [39]. Similarly, increased environmental carbon dioxide generally leads to

# Box 1. Analysis of Coevolution during Climate Change

To analyze the effects on densities and species traits that climate-driven changes in intrinsic rates of increase can have, we used an analytical approach akin to loop analysis [82]. This approach is complementary to the simulations used in the text and provides more general results that do not depend on the details of simulation models. As with loop analysis, we focus on changes in equilibrium densities with respect to changes in intrinsic rates of increase:

$$\frac{\partial X^*}{\partial E} = -\left(\frac{\partial G}{\partial X}\right)^{-1} \frac{\partial G}{\partial E},\tag{B1}$$

where  $X^* = (N_1^*, N_2^*, \upsilon_1^*, \upsilon_2^*)$  is the vector of the equilibrium densities and trait values for each species, *G* is a vector of functions that all equal zero when population densities and traits are at their equilibrium values (derived, for example, from Equations 1 and 2), and  $\partial G/\partial X$  is a matrix of derivatives of *G* with respect to *X*; thus,  $\partial G/\partial X = \mathbf{A}$ , where **A** is a 4×4 matrix of derivatives. Assuming that environmental change *E* affects only species 1, the derivative of the equilibrium density of species 1,  $N_1^*$ , with respect to *E* is proportional to -cofactor(**A**,1,1)/det(**A**), where cofactor(**A**,1,1) is the determinant of matrix **A** after the first row and first column are removed.

For the cases of competition and mutualism, we can use Equation B1 to analyze the effects of conflicting versus nonconflicting evolutionary interests encapsulated in the term *d*. Formally, *d* is a partial derivative giving the change in per capita interaction strength between species with respect to the change in the other species' trait value (Text S1). Nonetheless, for simplicity we represent this partial derivative as a single term *d* that we assume is the same for both species. When *d* is small ( $d \rightarrow 0$ ), the change in  $N_1^*$  with respect to *E* is:

$$\frac{\partial N_1^*}{\partial E} = \frac{C_1 + a_{44}a_{32}a_{23}d}{C_2 + a_{44}a_{32}a_{23}d + a_{33}a_{41}a_{14}d}\frac{\partial G_1}{\partial E},\qquad(B2)$$

where  $a_{ij}$  is the *ij*th element of the matrix **A**. The values  $C_1$ and  $C_2$  are positive constants such that in the absence of coevolution,  $\frac{\partial N_1^*}{\partial E} = \frac{C_1}{C_2} \frac{\partial G_1}{\partial E}$ . Because for competition and increased plant growth, although the strength of this effect varies from species to species [40]. In addition to broad-scale climatic changes such as these, our models have implications for environmental changes on a more local scale. For example, increased nitrogen and phosphorus runoff and land management regimes can each alter growth or mortality rates, and significantly degrade the structure of ecological communities [41–43]. We intentionally did not specify a particular type of environmental effect in order to retain the general applicability of the models, although we recognize that there are a myriad of different effects that environmental changes can bring, and climate change will likely affect multiple environmental factors that will directly impact species' population growth rates.

A key issue in our models is how changes in the trait value of one species affects the fitness of the other species. For example, suppose that selection on the trait of a competitor to decrease the strength of competition it experiences simultaneously decreased the strength of competition experienced by the second species. This could occur if

mutualism  $(1-\alpha^2)C_1 = C_2$ , it follows that  $C_1 > C_2$ . Combining this with the facts that  $a_{23}a_{32} > 0$ ,  $a_{14}a_{41} > 0$ ,  $a_{33} < 0$ , and  $a_{44} < 0$ , positive values of *d* increase  $\partial N_1 * / \partial E$ , whereas negative values of *d* decrease  $\partial N_1 * / \partial E$ . This shows that nonconflicting coevolution amplifies while conflicting coevolution dampens the effects of changing climate on population density  $N_1$ \* for any equation of the general form used for simulations (Equations 1 and 2).

In addition to generalizing the results from the simulations, approximation (B2) identifies the positive and negative feedback loops underlying the effects of coevolution, which are given by strings of  $a_{ii}$ 's. The first coevolutionary loop containing  $a_{32}a_{23}$  links the evolution of the trait of species 1 with changes in the density of species 2. The second coevolutionary feedback loop containing  $a_{41}a_{14}$ links the effect of evolution of the trait of species 2 to changes in the density of species 1. Thus, the effects of evolution explicitly involve coevolution of a species trait in response to changes in the density of the interacting species. This is emphasized by the result of the approximation that if the evolution of one species has no direct effect on the density of the other (i.e., d=0), evolution does not affect climate-driven changes in species densities. In other words, even if species evolve in response to climate change, and even if this evolutionary response changes the impact they experience from other species, this is not sufficient for evolution to change the response of their abundance to climate change. In addition, it is necessary for evolution of a species to affect its impact on other species it interacts with. When environmental change E affects both species, the derivative of the equilibrium density of species 1,  $N_1^*$ , with respect to E is equal to the sum of the derivative when only species 1 is affected and the derivative when only species 2 is affected. Thus, understanding the case where both species are affected by climate change can be easily determined by combining the cases where a single species is affected.

Finally, the same approximation approach applied to predator-prey coevolution gives the same results as for the competition and mutualism cases with d<0 for conflicting coevolution (Text S1).

the trait reduced competition by reducing the feeding niche overlap between competitors, so the second species would benefit from selection on the first. We refer to this case as nonconflicting coevolution. Conversely, if the trait were to make the first competitor more aggressive and hence better able to defend itself against the second competitor, then the second competitor would suffer from the evolution of the first. We refer to this as conflicting coevolution. As we will show, the consequences of coevolution for the abundance of species depend on whether changes in the trait of one species is beneficial or detrimental to its interacting partner that is, whether coevolution is nonconflicting or conflicting.

Competitors and mutualistic partners could experience either conflicting or nonconflicting coevolution, and different types of models have been used to describe each coevolutionary pathway. For example, competition models where competitors can reduce competition by shifting traits away from competitors (e.g., [35]) assume nonconflicting coevolution. In contrast, models focused on competitive arms races (e.g., [44]) assume conflicting coevolution between competitors. Although coevolution of mutualists is traditionally modeled as nonconflicting (e.g., [45]), we might expect conflicting coevolution to be common in mutualists as well. For example, yucca moths pollinate yucca plants while ovipositing in yucca flowers, and evolution of increased egg production within each flower leads to greater benefit received by the yucca moth, while negatively impacting yucca plants [46]. Thus, conflicting coevolution will occur for mutualists whenever there is the possibility of one partner cheating and reducing the benefit it provides [7,46,47]. For predator-prey interactions, evolution of prey to decrease the predation rate will generally be detrimental to the predator, whereas evolution of the predator to increase the predation rate will likely be detrimental to the prey. Therefore, coevolution of predator-prev interactions will generally be comparable to conflicting types of competition and mutualism, although as we discuss later, this might not strictly be the case for host-pathogen interactions.

To illustrate our theoretical results that are shared by all interactions—competition, mutualism, and predation—we used simple simulation models that share the characteristics discussed above. To aid the illustration, we selected parameter values intentionally to give coexistence of species (at least under some environmental conditions) and simple dynamics with stable equilibrium points. A theoretically more general, yet conceptually more challenging, approach to the same type of model is presented in Box 1; this general approach confirms that the qualitative patterns illustrated by our simulations are in fact found much more broadly under the general assumptions we have described.

# Competition

We modeled coevolution of two competitors using a discretetime, modified Lotka-Volterra competition model. The density of species i at time t,  $N_{i,t}$ , is given by

$$N_{i,t+1} = N_{i,t}F_i[r_i(E, v_{i,t}), N_{i,t} + \alpha_i(v_{i,t} + dv_{j,t})N_{j,t}]$$
  
=  $N_{i,t}e^{r_i(E, v_{i,t}) - N_{i,t} - \alpha_i(v_{i,t} + dv_{j,t})N_{j,t}},$  (1)

in which  $F_i$  gives the per capita population growth rate or, equivalently, the fitness of species *i*. The trait values that govern the strength of competition experienced by each species at time *t* are denoted  $v_{i,t}$  and  $v_{j,t}$ . The parameter  $\alpha_i(v_{i,t}+dv_{j,t})$  is the competition coefficient measuring the effect of species *j* on species *i*. We assumed  $\alpha_i(v_{i,t}+dv_{j,t}) = \exp(-v_{i,t}-dv_{j,t})$ . The parameter *d* determines whether coevolution is conflicting or nonconflicting, and hence is key to the model. If d < 0, then increases in  $v_{j,t}$  (which reduces competition experienced by species *j*) increases competition experienced by species *i*, thereby giving conflicting coevolution. Conversely, if d>0, then increases in  $\upsilon_{j,t}$  decrease competition experienced by species *i*, leading to nonconflicting coevolution. For simplicity, we assumed that the value of *d* is the same for both  $\alpha_1(\upsilon_{1,t}+d\upsilon_{2,t})$  and  $\alpha_2(\upsilon_{2,t}+d\upsilon_{1,t})$ , so that evolution has symmetric effects on both species.

The trait value  $v_{i,t}$  affects not only competition experienced by species *i* but also its intrinsic rate of increase  $r_i(E, v_{i,i})$ . Specifically, we assumed that  $r_i(E, v_{i,t}) = R_i + b_i E - f v_{i,t}$ , where f describes the cost of increasing  $v_{i,i}$ ; thus, there is a trade-off between reducing competition by increasing  $v_{i,t}$  in  $\alpha_i(v_{i,t}+dv_{i,t})$  and reducing the intrinsic rate of increase,  $r_i(E, v_{i,t})$ . Because we assumed that  $\alpha_i(v_{i,t}+dv_{i,t})$  has an exponential form, there is the possibility for an optimal fitness to be achieved at intermediate values of  $v_{i,t}$ . Other forms for  $\alpha_i(\upsilon_{i,t}+d\upsilon_{i,t})$  may lead to optimal fitness at either zero or infinite values of  $v_{i,i}$ ; we did not consider this situation, however, because these traits experiencing disruptive selection will likely fix within a local population. Finally, we assumed that the unspecified environmental variable E enhances the intrinsic rate of increase of species 1  $(b_1 > 0)$ , implying that E represents a more-favorable environment. For species 2, we assumed there is no effect of environmental change  $(b_2 = 0)$ .

In the model,  $\upsilon_{i,t}$  gives the mean value of a quantitative genetic trait whose distribution among individuals in the population is symmetric with additive genetic variance  $V_i$ . Provided the magnitude of the variance is not too large [38,48,49], selection for changes in the mean value  $\upsilon_{i,t}$  is equal to the derivative of fitness with respect to the trait divided by mean fitness [50]. For our model:

$$v_{i,t+1} = v_{i,t} + \frac{1}{F_i} \frac{\partial F_i}{\partial v_i} V_i = v_{i,t} + \left( N_{j,t} \frac{\partial \alpha_i (v_{i,t} + dv_{j,t})}{\partial v_i} - f \right) V_i.$$
(2)

# Mutualism

The model for two mutualists has the same structure as the competition model (1). For mutualism, the coefficient  $\alpha_i(v_{i,t}, v_{j,t}) = -\log(1+v_{i,t}+dv_{j,t})$  is negative, and the logarithmic form allows the optimal fitness to be achieved at intermediate values of  $v_{i,t}$ . As in the competition model, *d* determines whether coevolution is conflicting or nonconflicting. The other components of the model are the same as described for the competition model, and evolutionary change is described by Equation 2.

# Predation

For predator-prey interactions we used a discrete-time version of a model in which the predator attack rate is determined by traits of both prey,  $v_{1,b}$  and predator,  $v_{2,t}$  [51]. Prey trait  $v_{1,t}$  represents antipredator defense behavior, whereas predator trait  $v_{2,t}$  represents the ability of the predator to overcome prey defenses. Changes in the densities of prey  $N_t$  and predator  $P_t$  are given by:

$$N_{t+1} = N_t F_N[r(E, v_{1,t}), N_t, a(E, v_{2,t}, v_{1,t})P_t]$$
  
=  $N_t e^{r(E, v_{1,t}) - N_t - P_t a(E, v_{2,t}, v_{1,t})}$ (3)

 $P_{t+1} = P_t F_P[P_t, a(E, v_{2,t}, v_{1,t}) N_t, m(v_{2,t})] = P_t e^{cN_t a(E, v_{2,t}, v_{1,t}) - m(v_{2,t})},$ 

where  $r(E, v_{1,l}) = R + b_n E - f v_{1,l}$  is the intrinsic rate of increase of the prey and depends on the environmental variable *E*, as in the

competition and mutualism models. The predation rate  $a(E, v_{2,t})$  $\upsilon_{1,t} = q(E)\exp(-\upsilon_{2,t} \upsilon_{1,t})$  depends on the environmental variable E and declines with increasing prey defense,  $v_{1,t}$ , or the predator's susceptibility to the prey defense,  $v_{2,t}$ . We considered two scenarios, one in which the predation rate  $q(E) = Q_0 + b_0 E$  increases linearly with  $E(b_p > 0)$  while prey growth rate is unaffected  $(b_p = 0)$ , and the other in which the predation rate remains constant  $(b_{p} = 0)$  while the prey intrinsic rate of increase rises with  $E(b_n \ge 0)$ . Although we assumed q(E) is independent of prev density for simplicity, preliminary analyses showed that incorporating a nonlinear type II functional response [52] does not qualitatively alter the results. The predator experiences a cost of trait  $v_{2,t}$  in the form of increased mortality; specifically,  $m(v_{2,t}) = m_0 + g/v_{2,t}$ , where g governs the cost to the predator of being able to overcome the prey defense. Finally, if  $V_N$ and  $V_P$  are the additive genetic variances for prey and predators, respectively, evolution is given by:

$$v_{1,t+1} = v_{1,t} + \frac{1}{F_N} \frac{\partial F_N}{\partial v_1} V_N = v_{1,t} + \left( -P_t \frac{\partial a(E, v_{2,t}, v_{1,t})}{\partial v_1} - f \right) V_N (4)$$

$$v_{2,t+1} = v_{2,t} + \frac{1}{F_P} \frac{\partial F_P}{\partial v_2} V_P = v_{2,t} + \left( cN_t \frac{\partial a(E, v_{2,t}, v_{1,t})}{\partial v_2} + \frac{g}{v_{2,t}^2} \right) V_P.$$

# Results

To illustrate the importance of coevolution-especially the contrast between conflicting and nonconflicting coevolution-for the response of populations to environmental changes, we conducted two types of simulations. For each type, we assumed that the populations begin at eco-evolutionary equilibrium (i.e., traits and densities are both at equilibrium), with identical genetic variances for the two species. For mutualism and competition models, the two species were initially identical in every way except in their response to environmental change. For the first type of simulation, we tracked the trajectories of population densities and traits through time as the intrinsic rate of increase of one of the species increases with the environment, E. We compared the trajectories for different levels of genetic variance, because the lower the genetic variance, the slower the rate of evolution. The second type of simulation involved evaluating how environmental changes alter the ecological and coevolutionary equilibriums. To find these equilibriums, after changing the environment we simulated the models for an additional 1,000 generations to allow population densities and trait values to stabilize. We did not find alternative stable states, and thus present the single equilibrium for each scenario. These two types of simulations proved to give the same conclusions, with the simulations of trajectories giving only one additional piece of information: that trait values and densities moved uniformly to the equilibriums given by the second type of simulations. The correspondence between the two types of simulations results from the fact that the level of genetic variance determines the rate of approach to equilibrium but does not alter the equilibrium itself, which is a joint optimization of fitness in each species. To avoid redundancy, we only present the trajectories for the conflicting competition case, and subsequently focus solely on the equilibrium simulations. We refer the reader to Box 1 for a full mathematical treatment that does not depend on the specific equations we used for the simulation models. Finally, although we only considered two interacting species here, we have found qualitatively similar results in simulations of larger communities (results not shown).

### Competition

To illustrate the competition model, we began by simulating the consequences of raising the environmental quality for species 1 (increasing E) through time while varying the rate of coevolution. When the additive genetic variances for the traits expressed by both species,  $V_1$  and  $V_2$ , are zero, evolution cannot occur, whereas increasing  $V_1$  and  $V_2$  increases the rate of evolution. For this illustration we assumed competition is conflicting. Increasing E increases the density of species 1 and decreases the density of species 2, yet allowing evolution moderates both effects (Figure 1A). As  $V_1$  and  $V_2$  increase, the rate of change of population densities and trait values more closely track their equilibrium values  $\mathcal{N}_i^*$  and  $\upsilon_i^*$ , that is, the values at which, for fixed E,  $N_{i,t+1} = N_{i,t}$  and  $v_{i,t+1} = v_{i,t}$  in Equations 1 and 2 (Figure 1A,B). The effects of coevolution are largely driven by changes in trait values for species 2, with less change in species 1 (Figure 1B). This occurs because species 2 evolves to invest heavily in the competitive arms race, limiting the decline in investment by species 1.

For conflicting competition (d < 0, Figure 2A,C), equilibrium species densities are less sensitive to environmental change when there is coevolution, whereas coevolution augments changes in species densities when there is nonconflicting competition  $(d \ge 0,$ Figure 2B,D). This occurs because an increase in the density of species 1 with environmental change leads to a decrease in the density of species 2. Because selection pressure is positively correlated with the density of the other species, species 1 experiences relatively less selection pressure from competition with species 2 compared to the selection pressure on species 2 from species 1. When competition is conflicting (Figure 2A,C), the decreased selection on species 1 is beneficial to species 2, which acts to limit the decline of the population of species 2 and hence the decline of its effect on species 1. Also, the increased selection on species 2 increases its per capita competitive effect on species 1. These two sources of selective pressures combine to help species 2 and, in turn, are detrimental to species 1. When competition is nonconflicting (Figure 2B,D), the converse occurs; the decreased selection on species 1 caused by low densities of species 2 increases the effect of competition on species 2, and the increased selection on species 2 decreases its per capita competition effect on species 1. This selective pressure benefits species 1, further increasing its density.

In summary, conflicting competition sets up coevolution as a negative feedback, because selection on one species to reduce competition increases its competitive effect on the other species. In contrast, nonconflicting competition sets up coevolution as a positive feedback, because selection to reduce the impact of competition on one species also reduces the impact of competition on the other (Box 1).

# Mutualism

As with competition, the effects of coevolution on mutualists depended on the type of coevolution. When there is conflicting mutualism (d<0, Figure 3A,C), coevolution diminishes the effects of environmental change on equilibrium densities, in contrast to the case of nonconflicting mutualism (d>0, Figure 3B,D). This effect occurs because the increase in the density of species 1 due to the environmental change increases selection pressure on species 2 for investment in the mutualism. When the mutualism is conflicting, this change is detrimental to species 1 and limits its increase, because the benefits of mutualism decrease with the investment of species 2 in the interaction. In contrast, in the case of nonconflicting mutualism, increased investment by species 2 is beneficial to species 1, further increasing the density of species 1.



Climatic effect on intrinsic rate of increase

# Predation

For competition and mutualism, interacting species might have either conflicting or nonconflicting coevolutionary feedbacks. In contrast, predator and prey interactions are generally expected to exhibit conflicting coevolution and hence generate negative coevolutionary feedbacks: prey coevolution of defenses that reduce predation will be detrimental to the predator, and predator coevolution to increase the predation rate will be detrimental to prey. To verify this expectation, we analyzed both the case in which climate change increases the prey intrinsic rate of increase and the case in which climate change increases the predation rate and hence the predator population growth rate.

When climate change enhances the prey intrinsic rate of increase, the resulting increase in prey density leads to increased predator density, and in the absence of coevolution the equilibrium predator density increases dramatically (Figure 4A). In contrast, the equilibrium predator density increases more slowly when predator and prey coevolve (Figure 4A). As with conflicting competition and mutualism, higher predator density strengthens selection pressure for prey investment in the coevolutionary arms race (Figure 4C). With increased prey investment, the predator density cannot increase as much due to heightened prey defense (Figure 4A,C).

When climate change increases the predation rate but has no effect on the prey intrinsic rate of increase, the predator density increases rapidly in the absence of coevolution, but this increase is slowed by coevolution (Figure 4B,D). Because prey selection pressure is positively correlated with predator density, prey evolve higher defensive trait values in the presence of higher predation rates, which in turn lowers the predation rate, increases prey density, and decreases predator density. Thus, coevolution sets up a negative feedback loop that reduces the decline in prey density and increase in predator density (Figure 4B). While these results pertain to specialist predators that have no other prey species, we found that coevolution also reduces the ecological effects of climate change in a model for generalist predators (Figure S1).

# Discussion

We have shown, using simple models, that coevolution may increase or decrease the effect of environmental change, depending on the form that coevolution takes between species. In cases where species have conflicting interests, coevolution reduces the effects of environmental change on densities, because coevolution acts as a negative feedback to the effects of environmental change. Conversely, when species have nonconflicting interests, coevolution sets up a positive feedback that increases the effects of environmental change on densities. Given these contrasts, is coevolution in nature likely to involve conflicting or nonconflicting interests of interacting species? Competitors and mutualists, in particular, have the potential to coevolve along either conflicting or nonconflicting pathways. Thus, determining the predominant type of coevolution will be critical to identifying the long-term effects of climate change on species.

Below, we first give brief discussions of classical studies and show that cases of both conflicting and nonconflicting coevolution are common. Therefore, no *a priori* prediction can be made for their relative importance when anticipating the effects of climate change. We then turn to coevolutionary studies that directly address climate change, using these to show how evidence can be obtained to make and test predictions about the coevolutionary effects on specific systems facing climate change.

lines) densities (A) and trait values (B) as the climate variable *E* increases from 0 to 5 over the course of 100 time steps. The model describes conflicting competition (*d*<0), and additive genetic variances range from 0 (light blue) to 0.75 (red). Trajectories began at an eco-evolutionary equilibrium, and densities are scaled relative to this equilibrium. The trait value for species *i* dictates the strength of competition felt by species *i* per capita of species *j*. The top *x*-axis represents the climatic effect on the intrinsic rate of increase of species 1, *b*<sub>1</sub>*E*. Parameter values used were:  $R_1 = R_2 = 0.1$ , d = -0.9,  $b_{n1} = 0.001$ ,  $b_{n2} = 0$ , and f = 0.045.

Figure 1. Competition density and trait trajectories. For

competition, trajectories of species 1 (solid lines) and species 2 (dashed

doi:10.1371/journal.pbio.1001685.g001

In summary, conflicting mutualism sets up coevolution as a negative feedback, whereas nonconflicting mutualism sets up a positive feedback (Box 1).



# Climatic effect on intrinsic rate of increase

**Figure 2. Competition equilibrium densities and traits.** Equilibrium population densities (A, B) and trait values (C, D) for two competing species at different climatic conditions. The intrinsic rate of increase of species 1 (solid lines) increases linearly with climate *E*, while the intrinsic rate of increase of species 2 (dashed lines) is unaffected. Results are for conflicting competition, d < 0 (A, C), and nonconflicting competition, d > 0 (B, D). Red lines give eco-evolutionary equilibriums assuming high genetic variation ( $V_1$ ,  $V_2 > 0$ ), whereas for blue lines there is no evolution ( $V_1 = V_2 = 0$ ). The trait value (shown on the *y*-axis of C, D) for species *i* dictates the strength of competition gase (A, C) parameter values used were:  $R_1 = R_2 = 0.1$ , d = -0.9,  $b_1 = 0.001$ ,  $b_2 = 0$ , and f = 0.45. For the nonconflicting case (B, D) parameter values used were as follows:  $R_1 = R_2 = 0.1$ , d = 0.5,  $b_1 = 0.01$ ,  $b_2 = 0$ , and f = 0.02.

doi:10.1371/journal.pbio.1001685.g002

# **Competitive Coevolution**

It has long been recognized that coevolution can lead to increased asymmetries in competitive abilities [15], which is the hallmark of conflicting coevolution. But the idea that competition drives partitioning of food sources is even older [53–55], and this is the hallmark of nonconflicting coevolution. The effects of climate change for specific competitors hinge on which type of coevolution occurs. Evidence suggests that both are common.

Laboratory experiments that evaluate the effect of competitive interactions on trait evolution for each species have documented both conflicting coevolution in flies [15] and nonconflicting coevolution in *E. coli* strains [56]. Furthermore, conflicting and nonconflicting coevolution are not mutually exclusive; *Colpoda* 



# **Figure 3. Mutualism equilibrium densities and traits.** Equilibrium population densities (A, B) and trait values (C, D) under changing climatic conditions for conflicting, d < 0 (A, C), and nonconflicting mutualists, d > 0 (B, D). The intrinsic rate of increase of species 1 (solid lines) increases linearly with the climate variable *E*, while the intrinsic rate of increase of species 2 (dashed lines) is unaffected. The *x*-axis represents the climatic effect on the intrinsic rate of increase of species 1, $b_1E$ . Red lines give eco-evolutionary equilibriums assuming high genetic variation ( $V_1$ , $V_2 > 0$ ), whereas for blue lines there is no evolution ( $V_1 = V_2 = 0$ ). The trait value for species *i* (shown on the *y*-axis of C, D) dictates the benefits of mutualism accrued by species *i* per capita of species *j*. When the intrinsic rate of increase of species 1 is high enough and species are allowed to coevolve, there is no equilibrium in the nonconflicting mutualism model (B, D), as the growth of each species is unbounded. For the conflicting case (A, C) parameter values used were: $R_1 = R_2 = 0.2$ , d = -0.9, $b_1 = 0.025$ , $b_2 = 0$ , and f = 0.16. For the nonconflicting case (B, D) parameter values used were as follows: $R_1 = R_2 = 0.2$ , d = -0.9, $b_1 = 0.01$ , $b_2 = 0$ , and f = 0.16.

doi:10.1371/journal.pbio.1001685.g003



# **Climatic effects**

**Figure 4. Predation equilibrium densities and traits.** Equilibrium prey (solid lines) and predator (dashed lines) densities (A, B) and trait values (C, D) for different climatic conditions. Densities are scaled to equilibrium at E=0. (A, C) The prey intrinsic rate of increase increases linearly with climate *E*, while the predation rate is unaffected. (B, D) The predation rate increases linearly with *E*, while the prey intrinsic rate of increase is unaffected. The *x*-axis for panels A and C represents the climatic effect on the intrinsic rate of increase of prey,  $b_n E$ , and the *x*-axis for panels B and D represents the climatic effect on the intrinsic rate of increase of prey,  $b_n E$ , and the *x*-axis for panels B and D represents the climatic effect on the predation rate,  $b_p E$ . Red lines give eco-evolutionary equilibrium assuming high genetic variation ( $V_1$ ,  $V_2 >> 0$ ), and blue lines give the case of no coevolution ( $V_1 = V_2 = 0$ ). Increases in either prey or predator trait values reduce per capita predation rate. Parameter values used were: R = 0.5,  $Q_0 = 2$ , c = 0.25, f = 0.04, g = 0.04, and  $m_0 = 0.005$ . Climate change effect parameters were either  $b_p = 0.2$  and  $b_n = 0$  (A, C) or  $b_p = 0$  and  $b_n = 0.02$  (B, D). doi:10.1371/journal.pbio.1001685.g004

protozoans with initially weak competitive abilities have been shown to evolve along both pathways [57]. While these types of experimental studies have the advantage of documenting coevolution as it happens, they are limited by the range of species and time scales that are amenable to experiments, and the magnitude of environmental heterogeneity that may affect coevolution [58,59].

Alternatively, field studies can be used to infer the prevalence of conflicting versus nonconflicting coevolution. Research focusing on character displacement in natural populations attempts to identify the effects of coevolutionary processes based on species' phenotypes in solitary and sympatric populations [60]. This approach has documented both conflicting [61,62] and nonconflicting coevolution [63–65].

# Mutualistic Coevolution

There is a rich theory describing the evolution of mutualisms [66,67]. Theoretical predictions often suggest that mutualistic interactions have the potential to break down into parasitic interactions [47,68,69]; this is an extreme form of conflicting interests between species. If mutualism breakdown into parasitism is common, then conflicting coevolution is likely, and this will likely diminish the effects of climate change.

Nonetheless, if mutualistic partners can enforce good behavior of their partners [68], then nonconflicting coevolution is expected. For example, the plant *Medicago truncatula* discriminately rewards the most beneficial mycorrhizal partners with more carbohydrates, and mycorrhizal partners form partnerships only with the roots that provide the most carbohydrates [7]. Thus, each partner constrains the selection pressure of the other to allow only nonconflicting coevolution. If nonconflicting coevolution is frequently imposed by mutualists, our results suggest that coevolution between mutualistic species will exaggerate, rather than diminish, the effects of climate change on species densities.

# Predator-Prey Coevolution

Conflicting coevolution is expected for most types of predatorprey or consumer-resource interactions, because increases in prey defenses will decrease benefits to predators, and increases in predator effectiveness will be detrimental to prey. Nonetheless, evolution of parasite virulence could be different [70,71]. The conventional wisdom is that parasites should evolve to be less virulent, because this will increase their transmission among hosts; parasites are not transmitted by dead hosts, at least not for long [72]. Nonetheless, this ignores, among other things, the relationship between the production of large numbers of propagules (that generally harms the host) and transmission rates, and moredetailed analyses generally predict evolution of parasite virulence to represent a balance between higher virulence caused by selection for production of propagules and lower virulence caused by selection for lengthening the transmission period [73]. Therefore, evolution of the parasite may be nonconflicting with the host, even at the same time evolution of the host to limit infection is conflicting with the parasite. In models describing this interaction (results not shown), we found that when climatic changes directly affect the parasite, coevolution in the host fuels a negative feedback loop that mitigates the effects of climate change. In contrast, in some cases when climatic changes directly affect the host, coevolution can lead to a positive feedback loop that exaggerates the effects of climate change on the host density. Thus, when there are both conflicting and nonconflicting coevolution, the ultimate outcome will be determined by whether the host or parasite experiences greater evolutionary change.

# Predicting the Effects of Climate Change

Given the widespread occurrences of both conflicting and nonconflicting coevolution in competition and mutualism, and to a lesser extent in predator-prey interactions, systems will have to be studied on a case-by-case basis to predict and test the role of coevolution in modifying the effects of climate change. This could be done either using experimental studies or taking advantage of naturally occurring environmental gradients.

An example of an experimental study is given by Lopez-Pascua and Buckling [74], who performed an environmental manipulation of bacterial productivity by altering nutrient concentrations in the growth media. They showed that increasing bacterial productivity increases the rate of coevolution between bacteria and phages. They proposed that this is due, in part, to increased selection pressure on the bacterial population in environments with high productivity (high intrinsic rates of bacterial increase). This increased selection stems from increased encounters with phages, as phages numerically respond to increased bacterial density. The phages then evolve greater infectivity in response to bacterial evolution. This explanation is consistent with our theoretical expectations for conflicting evolution of prey and predators; increasing the prey intrinsic rate of increase leads to evolution of stronger prey defenses against the predator (Figure 4C).

In addition to experimental manipulations of environmental factors, it is possible to take advantage of natural environmental gradients similar to classical studies of character displacement. For example, in a field experiment, Toju et al. [75] documented a climatic gradient in a coevolutionary arms race between the camellia beetle (Curculio camelliae) and its host plant, Japanese camellia (Camellia japonica). Female beetles use their snout to pierce the camellia fruit pericarp and oviposit eggs into seeds, with oviposition success determined by the length of the beetle's snout and ovipositor relative to the pericarp thickness. Thus, plant defense is determined by pericarp thickness, and beetle snout and ovipositor lengths determine beetle ability to overcome this defense. The authors measured beetle and plant traits along a latitudinal gradient, and previous work had showed that plants exhibit faster potential for growth at lower latitudes [76]. Our analyses suggest that, because increases in prey growth should increase predator densities and, in turn, increase selection pressure on prey, the coevolutionary arms race should be "won" by prey under environmental conditions that favor prey population growth (Figure 4A,C). Thus, in the camellia-beetle arms race we expect that coevolution will favor plants more at lower latitudes. The authors indeed found this to be the case; plants in high latitude populations that experienced endemic predation by beetles had pericarp thicknesses similar to populations that did not experience beetles. In contrast, at lower latitudes plant populations that experienced beetle predation had thicker pericarps than populations that did not. There was thus an increase in plant defense along the environmental gradient. Furthermore, this plant defense increased with decreasing latitude at a greater rate than weevil ovipositor length, suggesting that plants exhibited a larger coevolutionary advantage in environmental conditions with increased prey growth [75]. These results support our theoretical predictions that higher prey intrinsic rates of increase should lead to a coevolutionary advantage to prey, thereby buffering the changes in predator densities driven by climate change.

The majority of coevolutionary studies involving environmental manipulations or environmental gradients have been conducted on predator-prey or herbivore-plant systems where conflicting coevolution is likely. Similar experiments that document changes in traits and density might help build a better understanding of coevolution in competitive and mutualistic relationships. Laboratory studies have suggested that coevolution can lead to a reversal of competitive hierarchy in just 24 generations [15], and can occur fast enough to drive population dynamics [16]. Therefore, experimental competition studies in which environmental factors are manipulated are possible for some types of organisms. Environmental gradient, rather than experimental, studies will be more practical for larger organisms with longer lifespans that operate at larger spatial scales. Using character displacement to infer conflicting versus nonconflicting coevolution is necessarily correlative, although it opens up the study of coevolution in the context of climate change to a much wider range of species under natural spatial and temporal scales.

Studies that evaluate coevolution over environmental gradients fit within the broader conceptual paradigm of geographic mosaic theory [77] in which differences in coevolutionary selection among spatially separated populations are analyzed as genotype by genotype by environment interactions. A key feature of geographic mosaic theory is that some local populations experience environmental conditions under which coevolutionary pressures are strong. These "coevolutionary hotspots" are characterized by fitness equations ([77], p. 100)

$$W_{1,E} = F_{1,E}(v_1, v_2)$$

$$W_{2,E} = F_{2,E}(v_1, v_2),$$
(5)

where the fitnesses  $W_{1,E}$  and  $W_{2,E}$  of species 1 and 2 depend on both phenotypes  $v_1$  and  $v_2$ , and on the environment *E*. This pair of equations has the same general structure as that we have used for Equations 1–4. Thus, our results address the possible character of evolution within coevolutionary hotspots, and how coevolutionary outcomes might differ under different environmental regimes.

We have only considered local populations, explicitly ignoring gene flow among populations. Thus, we have ignored the large body of theoretical and empirical studies evaluating gene flow among populations under different selective forces [77-79]. For example, Nuismer et al. [80] used spatially explicit population genetics models to show that isolated populations of mutualistic species were likely to reach equilibrium quickly, while antagonistic populations were likely to oscillate in both density and phenotype. When interaction types vary spatially, however, both dynamic and equilibrium clines occur, and the presence of each depends on the levels of selection and gene flow across the landscape [80]. In an experimental bacteria-bacteriophage community, bacteriophages became locally maladapted in the absence of gene flow, but became locally adapted when gene flow occurred between bacteriophage populations [81]. The importance of gene flow in both theoretical and empirical studies gives a caution to our recommendation that natural environmental gradients be used to assess the character of coevolution-conflicting versus nonconflicting-and whether coevolution sets up positive or negative feedback loops to environmental changes. Gene flow and a geographic mosaic of selective pressure may dampen or otherwise modify the effects of local selection on coevolutionary traits.

# References

- Bertrand R, Lenoir J, Piedallu C, Riofrio-Dillon G, de Ruffray P, et al. (2011) Changes in plant community composition lag behind climate warming in lowland forests. Nature 479: 517–520.
- Elith J, Graham CH, Anderson RP, Dudik M, Ferrier S, et al. (2006) Novel methods improve prediction of species' distributions from occurrence data. Ecography 29: 129–151.

# Conclusions

While it is recognized that evolution will play a role in determining how climatic changes directly affect species [18], the interactions among species force us to also consider coevolution between species. Our models suggest that the effects of coevolution on population densities depend on the presence of conflicting versus nonconflicting coevolutionary interests. While we encourage future studies that experimentally manipulate both coevolution and environmental change, we acknowledge that experiments are likely to be difficult logistically for most study systems. It may be possible, however, to use character displacement across environmental gradients to distinguish whether conflicting versus nonconflicting coevolution is impossible.

Experimental [15] and environmental gradient [60] approaches to infer the nature of coevolution are both five decades old, and we hope that our theoretical results provide new impetus for these types of studies. They give needed information to anticipate whether coevolution will increase or decrease the effects of climate change on the densities of interacting species.

# **Supporting Information**

Figure S1 Generalist predator equilibrium densities and traits. Equilibrium values of prey and generalist predator population densities (A, B) and traits (C, D) for different climatic conditions. Densities are scaled to the prey equilibrium density at E=0. (A, C) The prey intrinsic rate of increase rose linearly with climate E, while the predation rate was unaffected. (B, D) The predation rate increased linearly with climate E, while prey growth was unaffected. Red lines give eco-evolutionary equilibrium assuming high genetic variation ( $V_1$ ,  $V_2 >>0$ ), and blue lines give the case of no coevolution ( $V_1 = V_2 = 0$ ). Parameter values used were:  $R_n = 0.5$ ,  $R_p = 0.2$ ,  $Q_0 = 2$ , c = 0.25, f = 0.04, g = 0.04, and  $m_0 = P_t - 0.2$  (to account logistic growth on alternative resources). Climate change effect parameters were either  $b_p = 0.2$  and  $b_n = 0$  (A, C), or  $b_p = 0$ and  $b_n = 0.02$  (B, D).

(TIF)

# Text S1 Analytical approximation for changes in species abundances with coevolution. (DOC)

### Acknowledgments

J.A. Rosenheim, B.T. Barton, H. Fan, C. Herren, B.A. Krimmel, M.H. Meisner, E.G. Murell, A. Sadeh, J. Usinowicz, K. Webert, and K.A. Zemenick provided helpful discussion.

# **Author Contributions**

The author(s) have made the following declarations about their contributions: Conceived and designed the experiments: TDN ARI. Performed the experiments: TDN ARI. Analyzed the data: TDN ARI. Wrote the paper: TDN ARI.

- Biesmeijer JC, Roberts SPM, Reemer M, Ohlemuller R, Edwards M, et al. (2006) Parallel declines in pollinators and insect-pollinated plants in Britain and the Netherlands. Science 313: 351–354.
- Tylianakis JM, Didham RK, Bascompte J, Wardle DA (2008) Global change and species interactions in terrestrial ecosystems. Ecol Lett 11: 1351– 1363.

- Pounds JA, Bustamante MR, Coloma LA, Consuegra JA, Fogden MPL, et al. (2006) Widespread amphibian extinctions from epidemic disease driven by global warming. Nature 439: 161–167.
- Harmon JP, Moran NA, Ives AR (2009) Species response to environmental change: impacts of food web interactions and evolution. Science 323: 1347–1350.
- Kiers ET, Palmer TM, Ives AR, Bruno JF, Bronstein JL (2010) Mutualisms in a changing world: an evolutionary perspective. Ecol Lett 13: 1459–1474.
- Araujo MB, Rozenfeld A, Rahbek C, Marquet PA (2011) Using species cooccurrence networks to assess the impacts of climate change. Ecography 34: 897–908.
- Barton BT (2011) Local adaptation to temperature conserves top-down control in a grassland food web. Proc R Soc Lond Ser B Biol Sci 278: 3102–3107.
- Jump AS, Penuelas J (2005) Running to stand still: adaptation and the response of plants to rapid climate change. Ecol Lett 8: 1010–1020.
- Bradshaw WE, Holzapfel CM (2006) Climate change evolutionary response to rapid climate change. Science 312: 1477–1478.
- Hairston NG, Ellner SP, Geber MA, Yoshida T, Fox JA (2005) Rapid evolution and the convergence of ecological and evolutionary time. Ecol Lett 8: 1114–1127.
- 13. Schoener TW (2011) The newest synthesis: understanding the interplay of evolutionary and ecological dynamics. Science 331: 426–429.
- Zhang QG, Buckling A (2011) Antagonistic coevolution limits population persistence of a virus in a thermally deteriorating environment. Ecol Lett 14: 282–288.
- Pimentel D, Feinberg EH, Wood PW, Hayes JT (1965) Selection, spatial distribution, and coexistence of competing fly species. Am Nat 99: 97–110.
- Yoshida T, Jones LE, Ellner SP, Fussmann GF, Hairston NG (2003) Rapid evolution drives ecological dynamics in a predator-prey system. Nature 424: 303–306.
- Becks L, Ellner SP, Jones LE, Hairston NG (2010) Reduction of adaptive genetic diversity radically alters eco-evolutionary community dynamics. Ecol Lett 13: 989–997.
- Hoffmann AA, Sgro CM (2011) Climate change and evolutionary adaptation. Nature 470: 479–485.
- Norberg J, Urban MC, Vellend M, Klausmeier CA, Loeuille N (2012) Ecoevolutionary responses of biodiversity to climate change. Nat Clim Change 2: 747–751.
- Kubisch A, Degen T, Hovestadt T, Poethke HJ (2013) Predicting range shifts under global change: the balance between local adaptation and dispersal. Ecography 36: 1–10.
- Hochberg ME, van Baalen M (1998) Antagonistic coevolution over productivity gradients. Am Nat 152: 620–634.
- Abrams PA, Vos M (2003) Adaptation, density dependence and the responses of trophic level abundances to mortality. Evol Ecol Res 5: 1113–1132.
- Friman VP, Hiltunen T, Laakso J, Kaitala V (2008) Availability of prey resources drives evolution of predator-prey interaction. Proc R Soc Lond Ser B Biol Sci 275: 1625–1633.
- Boots M (2011) The evolution of resistance to a parasite is determined by resources. Am Nat 178: 214–220.
- Tien RJ, Ellner SP (2012) Variable cost of prey defense and coevolution in predator-prey systems. Ecol Monogr 82: 491–504.
- Kraaijeveld AR, Godfray HCJ (1997) Trade-off between parasitoid resistance and larval competitive ability in *Drosophila melanogaster*. Nature 389: 278–280.
- Brodie ED, Brodie ED (1999) Costs of exploiting poisonous prey: evolutionary trade-offs in a predator-prey arms rage. Evolution 53: 626–631.
- Yoshida T, Hairston NG, Ellner SP (2004) Evolutionary trade-off between defence against grazing and competitive ability in a simple unicellular alga, *Chlorelia vulgaris.* Proc R Soc Lond Ser B Biol Sci 271: 1947–1953.
- Abrams PA (1987) Alternative models of character displacement and niche shift.
   displacement when there is competition for a single resource. Am Nat 130: 271–282.
- Case TJ, Taper ML (2000) Interspecific competition, environmental gradients, gene flow, and the coevolution of species' borders. Am Nat 155: 583–605.
- Pimentel D (1968) Population regulation and genetic feedback. Science 159: 1432–1437.
- Pimentel D, Levin SA, Soans AB (1975) Evolution of energy-balance in some exploiter-victim systems. Ecology 56: 381–390.
- Kawecki TJ, Ebert D (2004) Conceptual issues in local adaptation. Ecol Lett 7: 1225–1241.
- Gomulkiewicz R, Thompson JN, Holt RD, Nuismer SL, Hochberg ME (2000) Hot spots, cold spots, and the geographic mosaic theory of coevolution. Am Nat 156: 156–174.
- Roughgarden J (1976) Resource partitioning among competing species a coevolutionary approach. Theor Popul Biol 9: 388–424.
- Jones EI, Gomulkiewicz R (2012) Biotic interactions, rapid evolution, and the establishment of introduced species. Am Nat 179: E28–E36.
- Nuismer SL, Doebeli M, Browning D (2005) The coevolutionary dynamics of antagonistic interactions mediated by quantitative traits with evolving variances. Evolution 59: 2073–2082.
- Abrams PA (2001) Modelling the adaptive dynamics of traits involved in interand intraspecific interactions: an assessment of three methods. Ecol Lett 4: 166– 175.
- Gillooly JF, Charnov EL, West GB, Savage VM, Brown JH (2002) Effects of size and temperature on developmental time. Nature 417: 70–73.

- Ainsworth EA, Long SP (2005) What have we learned from 15 years of free-air CO2 enrichment (FACE)? A meta-analytic review of the responses of photosynthesis, canopy. New Phytol 165: 351–371.
- Tylianakis JM, Tscharntke T, Lewis OT (2007) Habitat modification alters the structure of tropical host-parasitoid food webs. Nature 445: 202–205.
- Carpenter SR, Caraco NF, Correll DL, Howarth RW, Sharpley AN, et al. (1998) Nonpoint pollution of surface waters with phosphorus and nitrogen. Ecol Appl 8: 559–568.
- Crowder DW, Northfield TD, Strand MR, Snyder WE (2010) Organic agriculture promotes evenness and natural pest control. Nature 466: 109–112.
- Abrams PA, Matsuda H (1994) The evolution of traits that determine ability in competitive contests. Evol Ecol 8: 667–666.
   Karry M, Gravillar S, (2006) Melcillarup contraits and the computing of
- Kopp M, Gavrilets S (2006) Multilocus genetics and the coevolution of quantitative traits. Evolution 60: 1321–1336.
- Pellmyr O, Huth CJ (1994) Evolutionary stability of mutualism between yuccas and yucca moths. Nature 372: 257–260.
- Thompson JN, Cunningham BM (2002) Geographic structure and dynamics of coevolutionary selection. Nature 417: 735–738.
- Iwasa Y (1991) Sex change evolution and cost of reproduction. Behav Ecol 2: 56–68.
- Abrams PA, Matsuda H, Harada Y (1993) Evolutionarily unstable fitness maxima and stable fitness minima of continuous traits. Evol Ecol 7: 465–487.
- Roff DA (1997) Evolutionary quantitative genetics: Chapman and Hall, Inc., New York 493 p.
- Ives AR, Dobson AP (1987) Antipredator behavior and the population-dynamics of simple predator-prey systems. Am Nat 130: 431–447.
- Holling CS (1959) The components of predation as revealed by a study of smallmammal predation of the European pine sawfly. Can Entomol 91: 293–320.
- Gause GF (1934) The struggle for existence: Balitmore, USA: Williams & Wilkins Co. 163 p.
- Lawlor LR, Smith JM (1976) Co-evolution and stability of competing species. Am Nat 110: 79–99.
- Schluter D (1994) Experimental-evidence that competition promotes divergence in adaptive radiation. Science 266: 798–801.
- 56. Tyerman JG, Bertrand M, Spencer CC, Doebeli M (2008) Experimental demonstration of ecological character displacement. BMC Evol Biol 8.
- 57. terHorst CP (2011) Experimental evolution of protozoan traits in response to interspecific competition. J Evol Biol 24: 36-46.
- Rainey PB, Travisano M (1998) Adaptive radiation in a heterogeneous environment. Nature 394: 69–72.
- Schoener TW (1974) Resource partitioning in ecological communities. Science 185: 27–39.
- 60. Brown WL, Wilson EO (1956) Character displacement. Syst Zool 5: 49-64.
- Adams DC (2004) Character displacement via aggressive interference in appalachian salamanders. Ecology 85: 2664–2670.
- Peiman KS, Robinson BW (2007) Heterospecific aggression and adaptive divergence in brook stickleback (*Culaea inconstans*). Evolution 61: 1327–1338.
- Schluter D, Mcphail JD (1992) Ecological character displacement and speciation in sticklebacks. Am Nat 140: 85–108.
- Schluter D, Price TD, Grant PR (1985) Ecological character displacement in Darwin finches. Science 227: 1056–1059.
- Schoener TW (1970) Size patterns in West Indian Anolis lizards. 2. correlations with sizes of particular sympatric species - displacement and convergence. Am Nat 104: 155–174.
- Bruno JF, Stachowicz JJ, Bertness MD (2003) Inclusion of facilitation into ecological theory. Trends Ecol Evol 18: 119–125.
- Bronstein JL (2009) The evolution of facilitation and mutualism. J Ecol 97: 1160–1170.
- Herre EA, Knowlton N, Mueller UG, Rehner SA (1999) The evolution of mutualisms: exploring the paths between conflict and cooperation. Trends Ecol Evol 14: 49–53.
- Wilson WG, Morris WF, Bronstein JL (2003) Coexistence of mutualists and exploiters on spatial landscapes. Ecol Monogr 73: 397–413.
- Alizon S, Hurford A, Mideo N, Van Baalen M (2009) Virulence evolution and the trade-off hypothesis: history, current state of affairs and the future. J Evol Biol 22: 245–259.
- Little TJ, Shuker DM, Colegrave N, Day T, Graham AL (2010) The coevolution of virulence: tolerance in perspective. Plos Pathog 6 (9): e1001006. doi:10.1371/ journal.ppat.1001006
- 72. Palmieri JR (1982) Be fair to parasites. Nature 298: 220-220.
- 73. May RM, Anderson RM (1983) Epidemiology and genetics in the coevolution of
- parasites and hosts. Proc R Soc Lond Ser B Biol Sci 219: 281–313.
  74. Lopez-Pascua LDC, Buckling A (2008) Increasing productivity accelerates host-parasite coevolution. J Evol Biol 21: 853–860.
- Toju H, Abe H, Ueno S, Miyazawa Y, Taniguchi F, et al. (2011) Climatic gradients of arms race coevolution. Am Nat 177: 562–573.
- Miyazawa Y, Kikuzawa K (2006) Physiological basis of seasonal trend in leaf photosynthesis of five evergreen broad-leaved species in a temperate deciduous forest. Tree Physiol 26: 249–256.
- Thompson JN (2005) The geographic mosaic of coevolution. Chicago: University of Chicago Press, 425 p.
- Forde SE, Thompson JN, Bohannan BJM (2004) Adaptation varies through space and time in a coevolving host-parasitoid interaction. Nature 431: 841–844.

- 79. Slatkin M (1987) Gene flow and geographic structure of natural populations. Science 236: 787-792.
- Nuismer SL, Thompson JN, Gomulkiewicz R (2000) Coevolutionary clines across selection mosaics. Evolution 54: 1102–1115.
- 81. Morgan AD, Gandon S, Buckling A (2005) The effect of migration on local
- adaptation in a coevolving host-parasite system. Nature 437: 253–256. 82. Levins R (1974) Qualitative analysis of partially specified systems. Ann NY Acad Sci 231: 123–138.